Author's response to reviews

Title: A case-control study of glycemic index, glycemic load and dietary fiber intake and risk of adenocarcinomas and squamous cell carcinomas of the esophagus: the Australian Cancer Study

Authors:

Petra H Lahmann (plahmann@gmx.de)
Torukiri I Ibiebele (Torukiri.Ibbiebele@qimrberghofer.edu.au)
Penelope M Webb (Penny.Webb@qimrberghofer.edu.au)
Christina M Nagle (Christina.Nagle@qimrberghofer.edu.au)
David C Whiteman (David.Whiteman@qimrberghofer.edu.au)

Version: 2 Date: 17 March 2014

Author's response to reviews:

Provision of line numbering